Author's response to reviews

Title: An unusual timing for symptomatic chest pain in adult chest wall myofibroma: a case report

Authors:

Chin-Li Chen (j0921713355@yahoo.com.tw)
Hung Chang (hungc888@yahoo.com.tw)

Version: 2 Date: 28 March 2014

Author's response to reviews: see over
Author's response to reviews

Title: Chest pain due to an unusual chest wall myofibroma in an adult: a case report

Authors:
Chin-Li Chen (j0921713355@yahoo.com.tw)
Hung Chang (hunge888@yahoo.com.tw)

Version: 2 Date: 24 March 2014

Author's response to reviews: see over
Reviewer's report
Title: Chest pain due to an unusual chest wall myofibroma in an adult: a case report
Version: 1 Date: 17 February 2014

Reviewer: Rakesh Garg

Has the case been reported coherently?: No

Is the case report authentic?: Yes

Is the case report ethical?: No

Is there any missing information that you think must be added before publication?: Yes

Is this case worth reporting?: No

Is the case report persuasive?: No

Does the case report have explanatory value?: No

Does the case report have diagnostic value?: No

Will the case report make a difference to clinical practice?: No

Is the anonymity of the patient protected?: Yes

Comments to authors:

The details of the procedures, pain, surgery has not been described in detail. The pain after any musculoskeletal cancer have pain. There is no new message from the case

Thank you very much for the comment. We have modified the title to “An unusual timing for symptomatic chest pain in adult chest wall myofibroma: a case report”, to make the rarity of this case more clear. Pain characteristics and procedure details were added to the Case presentation of our manuscript.

Declaration of competing interests:
I declare that I have no competing interests

Reviewer's report
Title: Chest pain due to an unusual chest wall myofibroma in an adult: a case report

Version: 1  Date: 10 March 2014

Reviewer: Tumul Chowdhury

Which of the following best describes what type of case report this is?:
Unexpected or unusual presentations of a disease

Thank you very much for the comment. We have modified the title to “An unusual timing for symptomatic chest pain in adult chest wall myofibroma: a case report”, to make the rarity of this case more clear.

Has the case been reported coherently?: Yes

Is the case report authentic?: Yes

Is the case report ethical?: Yes

Is there any missing information that you think must be added before publication?: No

Is this case worth reporting?: Yes

Is the case report persuasive?: No

Does the case report have explanatory value?: Yes

Does the case report have diagnostic value?: Yes

Will the case report make a difference to clinical practice?: No

Is the anonymity of the patient protected?: Yes

Comments to authors:

Dear Authors,

This is a rare but interesting report. I have few minor comments
Delete these lines in discussion,(this is a rare report,.....,)it is repeated two three times. It will be good if you have a table showing differential diagnosis of chest wall tumours and chest pain association...

Thank you very much for the comment.

1. The repeated verses were now deleted, making the manuscript more concise.

2. We have made a simplified table to show common chest wall tumors and the associated symptoms. Please refer to Table 1.

**Quality of written English:** Acceptable

**Declaration of competing interests:**

I declare that I have no competing interests